Author’s response to reviews

Title: Randomized clinical trial of efficacy and safety of colchicine 0.5% cream versus photodynamic therapy with methyl aminolevulinate in treatment of skin field cancerization: study protocol

Authors:
Anna Miola (anna_fmrp@yahoo.com.br)
Eliane Roio (eliane_roio@yahoo.com.br)
Helio Miot (heliomiot@gmail.com)
Luciana Abbade (lfabbade@fmb.unesp.br)
Juliano Schimitt (julivs@gmail.com)

Version: 2 Date: 20 Nov 2016

Author’s response to reviews:

Dear editor,

Thank you for the answer about the Study Protocol "Randomized clinical trial of efficacy and safety of colchicine 0.5% cream versus photodynamic therapy with methyl aminolevulinate in treatment of skin field cancerization: study protocol"

According to items requested:

"1) Ethical and Funding Approval Documentation:

Before we can proceed with your submission, can you please forward copies of all ethical approval and funding approval for our records. These documents should be sent as email attachments to the following email address, BMCSeriesEditorial@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system."
The requested document was sent to BMCSeriesEditorial@biomedcentral.com.

Best wishes

Anna Carolina Miola